\begin{table}%t1
\caption{\label{tab1}}
%\centering
\par
\begin{tabular}{ccccccccccc}
\hline\hline
 $N$ & Name &  $P$ & $q$ & $M_{1}$ & $R_{1}$ & $M_{bol_{1}}$ &
$r_{1}$ &
 $L_{1}$ & $i$ & Ref. \\
  &      & $S\!p$ & $A$ & $M_{2}$ & $R_{2}$ & $M_{bol_{2}}$ & $r_{2}$ &
$J_{1}/J_{2}$ &
$W$ & \\
\hline 1 & $RT$  & $0\fd62892942^{\star}$ & $\underline{0.74}$ &
1.23 & 1.25 & 4.04 & 0.315
& 0.886 {\it V} & $88\fdg4$ & {\footnotesize Arevalo et~al. (\cite{Arevalo}) }\\
  & $And$ &   $ F9 {\it V} + K2.5 {\it V} $   & 3.98 & 0.91 & 0.90 & 5.73 & 0.227 &  4.05 {\it V} &
9 & {\footnotesize Popper (\cite{Popper94}) } \\
  &  &   &   &  &  &  &  &  &  & {\footnotesize Kjurkchieva et~al. (\cite{Kjurkchieva}) }\\
\hline
  &    & $0\fd695662240^{\star}$ & 0.79 & 1.27 & 1.91 & 2.80 & 0.440 & 0.76  {\it V} &
$80\fdg8$ & {\footnotesize Lavrov (\cite{Lavrov}) }\\
2 & $WZ$ &    $ F5 {\it V} + [G6]{\it V} $  & 4.34 & 1.00 & 1.43 & 3.96 &
0.329 & 1.77  {\it V} &
4 & {\footnotesize Cook (\cite{Cook}) }\\
\cline{3-11}
  & $And$ & $0\fd695660340^{\star}$ & 0.67 & 2.10 & 2.31 & 2.39 & 0.461 & 0.81  {\it V} &
$81\fdg8$ & {\footnotesize Kyu-Dong (\cite{Kyu-Dong}) }\\
  &    &  $ F5 {\it V} + [G8.5]{\it V} $  & 5.01 & 1.40 & 1.72 & 3.80 & 0.344 & 2.37  {\it V} &
4 & \\
\hline 3 & $BX$ & $0\fd61011355^{\star}$ & $\underline{0.49}$ &
1.52 & 1.78 & 2.79 & 0.448
& 0.94 {\it V} & $74\fdg6$ & {\footnotesize Bell et~al. (\cite{Bell}) }\\
  & $And$ &  $ F3 {\it V} + [K4] {\it V} $  & 3.98 & 0.75 & 1.30 & 5.26 & 0.326 & 8.3  {\it V} &
7 & {\footnotesize Derman et~al. (\cite{Derman}) } \\
  &  &   &   &  &  &  &  &  &  & {\footnotesize  Jassur et~al. (\cite{Jassur}) }\\
\hline 4 & $DS$ & $1\fd01051872$      & $\underline{0.59}$ &
1.58 & 2.10 & 2.39 & 0.365
& 0.85 {\it V} & $84\fdg3$ & {\footnotesize Schiller \& Milone (\cite{Schiller}) }\\
  & $And$ &  $ F3 I{\it V}-{\it V} + G0 {\it V} $   & 5.76 & 0.94 & 1.19 & 4.15 & 0.207 & 1.82 {\it V} &
7 & \\
\hline 5 & $HS$  & $0\fd71016230^{\star}$ & $\underline{0.64}$ &
1.35 & 1.35 & 3.88 & 0.31
& 0.77 {\it V} & $68^{\circ}$ & {\footnotesize Robb et~al. (\cite{Robb}) }\\
  & $Aqr$ & $ F9 {\it V} + [K2.5] {\it V} $ & 4.36 & 0.86 & 1.35 & 4.84 & 0.31  & 3.35 {\it V} &
4 & {\footnotesize Marschall et~al. (\cite{Marschall}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize  Popper (\cite{Popper95}) }\\
\hline 6 & $RX$  & $1\fd02962682$      & 0.27 & 1.30 & 1.96 &
2.52 & 0.386 & 0.99 {\it V} &
$81\fdg9$ & {\footnotesize Wilson \&  Rafert (\cite{Wilson}) }\\
  & $Ari$ &  $ F2 {\it V} + [M0] {\it V} $  & 5.08 & 0.36 & 1.10 & 6.28 & 0.216 & 31   {\it V} &
4 & {\footnotesize Kreiner et~al. (\cite{Kreiner}) }\\
\hline 7 & $C{\it V}$  &  $0\fd8469935$      & $\underline{0.94}$ &
1.02 & 1.12 & 4.40 & 0.236
& 0.56 {\it V} & $83^{\circ}$ & {\footnotesize Popper (\cite{Popper93}) }\\
  & $Boo$ &  $G3: {\it V} + [G6] {\it V} $  & 4.73 & 0.96 & 1.05 & 4.64 & 0.221 & 1.11 {\it V} &
5 & {\footnotesize Busch (\cite{Busch}) }\\
\hline 8 & $S{\it V}$  & $0\fd5930718^{\star}$  & $\underline{0.59}$ &
1.47 & 1.29 & 3.67 & 0.328
& 0.96 {\it V} & $90^{\circ}$ & {\footnotesize Patkos \&  Hempelmann (\cite{Patkos}) }\\
  & $Cam$ & $ F5 {\it V} + [K4.5] {\it V} $ & 3.94 & 0.87 & 0.83 & 6.30 & 0.211 & 9.93 {\it V} &
9 & {\footnotesize Pojmanski (\cite{Pojmanski}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize Frigo et~al. (\cite{Frigo}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize Kjurkchieva et~al. (\cite{Kjurkchieva02}) }\\
\hline 9 & $WX$  &  $1\fd2245888$  & $\underline{0.74}$ & 1.31 &
1.51 & 3.19 & 0.238
& 0.73 {\it V} & $82\fdg6$ & {\footnotesize Lu et~al. (\cite{Lu90}, \cite{Lu}) }\\
  & $Cnc$ &    $ F2 {\it V} + G4 {\it V} $    & 6.34 & 0.97 & 1.28 & 4.28 & 0.202 & 1.95 {\it V} &
7 &  \\
\hline 10 & $WY$  &  $0\fd82937122$    & $\underline{0.51}$ &
1.17 & 1.07 & 4.54 & 0.239
& 0.97 {\it V} & $87\fdg8$ & {\footnotesize Pojmanski (\cite{Pojmanski}) }\\
  & $Cnc$ &    $ G5 {\it V} + K2 {\it V} $    & 4.49 & 0.60 & 0.67 & 6.28 & 0.150 & 12.7 {\it V} &
8 & {\footnotesize Heckert (\cite{Heckert03}) }\\
\hline 11 & ${\it V}Z$  &  $0\fd84246163$   & $\underline{0.78}$ &
1.84 & 1.76 & 2.68 & 0.315
& 0.78 {\it V} & $80^{\circ}$ & {\footnotesize Popper (\cite{Popper88}) }\\
  & $C{\it V}n$ & $ F2 {\it V} + [F8] {\it V} $  & 5.57 & 1.43 & 1.25 & 3.98 & 0.225 & 1.80 {\it V} &
8 & {\footnotesize Shenghong et~al. (\cite{Shenghong}) }\\
\hline 12 & $AE$  &  $0\fd7591215$    & 0.7: & 1.0: & 1.50 &
3.93 & 0.356 & 0.82 {\it V} &
$80\fdg4$ & {\footnotesize Srivastava \&  Kandpal (\cite{Srivastava84}) }\\
  & $Cas$ & $(G6-7) {\it V}+[K4.5] {\it V}$ & 4.2 & 0.7: & 1.45 & 5.07 & 0.345 & 4.28 {\it V} &
2 & {\footnotesize Nelson (\cite{Nelson}) }\\
\hline 13 & $ST$  &  $1\fd2234184$    & 0.90 & 1.13 & 2.02 &
2.85 & 0.325 & 0.52 {\it V} &
$83\fdg7$ & {\footnotesize Milone \&  Hrivnak (\cite{Milone}) }\\
   & Cen &  $ F 9 {\it V} + F 9 {\it V} $  & 6.21 & 1.02 & 1.97 & 2.94 & 0.317 & 1.03 {\it V} &
3 & \\
\hline 14 & $RS$  &  $0\fd67235579$   & 1:   & 1.1  & 1.2  &
4.17 & 0.28  & 0.53 {\it V} &
$67\fdg6$ & {\footnotesize McFarlane \&  Hilditch (\cite{McFarlane}) }\\
   & $Col$ &  $G0 {\it V} + [G4] {\it V}$  & 4.2  & 1.1  & 1.2  & 4.28 & 0.28  & 1.13 {\it V} &
2 & {\footnotesize Rucinski (\cite{Rucinski}) }\\
\hline  15 & $CG$  &  $0\fd63114100$   & $\underline{0.84}$ &
0.96 & 0.89 & 5.36 & 0.239
& 0.70 {\it V} & $82\fdg4$ & {\footnotesize Popper (\cite{Popper94}) }\\
   & $Cyg$ &  $ G9.5 {\it V} + K3 {\it V} $  & 3.74 & 0.81 & 0.84 & 5.98 & 0.224 & 2.09 {\it V} &
7 & {\footnotesize Dapergolas et~al. (\cite{Dapergolas}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize Kjurkchieva et~al. (\cite{Kjurkchieva03}) }\\
\hline 16 & $AR$  &  $0\fd67583743$  & 0.75: & 1.27 & 1.49 &
3.34 & 0.354 & 0.82 {\it V} &
$84\fdg5$ & {\footnotesize Broglia \&  Conconi (\cite{Broglia}) }\\
   & $Dra$ &  $F5 {\it V} + [G9] {\it V}$ & 4.22  & 0.95 & 1.32 & 4.62 & 0.314 & 3.2 {\it V} &
3 & \\
\hline 17 & $BZ$ & $0\fd6641703^{\star}$ & 0.93 & 1.0  & 1.1  & 4.5
& 0.28  & 0.63 {\it V} &
$69^{\circ}$ & {\footnotesize Srivastava \&  Sinha (\cite{Srivastava}) }\\
   & $Eri$ & $(G0 III)+(G2 III)$ & 4.0  & 0.93 & 1.0  & 5.0  & 0.25  & 1.4  {\it V} &
1 & {\footnotesize Srivastava \&  Uddin (\cite{Srivastava86}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize  Srivastava (\cite{Srivastava1986}) }\\
\hline 18 & $MT$  &  $0\fd48771779$   & 0.58 & 1.2: & 1.0: & 4.5
& 0.303 & 0.94 {\it V} &
$77\fdg6$ & {\footnotesize Budding \&  Murad (\cite{Budding89}) }\\
   & $Her$ &  $F8 {\it V} + [K6] {\it V}$  & 3.2  & 0.7: & 0.7: & 6.8  & 0.222 & 8.4 {\it V} &
3 & \\
\hline 19 & $CN$  &  $0\fd637359955$  & 0.77 & 1.07 & 1.31 & 4.06 &
0.34  & 0.86 {\it V} &
$78\fdg1$ & {\footnotesize Nekrasova (\cite{Nekrasova}) }\\
   & $Lac$ & $G3 {\it V} + [K1.5] {\it V}$ & 3.85 & 0.82 & 0.81 & 5.80 & 0.21  & 2.37 {\it V} &
2 & {\footnotesize Kreiner (\cite{Kreiner}); CDS }\\
\hline
   &     &  $0\fd600086414$  & $\underline{0.97}$ & 1.13 & 1.17 & 4.29 & 0.301
& 0.53  {\it V} & $83\fdg2$ & {\footnotesize Botsula (\cite{Botsula}); Popper (\cite{Popper93}) }\\
 20 & $U{\it V}$  &  $ G 2 {\it V} + G 2 {\it V} $  & 3.90 & 1.09 & 1.13 & 4.36 &
0.290 & 1.05  {\it V} &
7--8 & {\footnotesize Popper (\cite{Popper95}); Wunder (\cite{Wunder}) }\\
\cline{3-11}
  & $Leo$ & $0\fd600086675^{\star}$ & $\underline{0.92}$ & 1.21 & 1.10 & 4.38 & 0.277
& 0.47 {\it V} & $82\fdg6$ & {\footnotesize Frederic \&  Etzel (\cite{Frederic}) }\\
   &     &  $ G O {\it V} + G 2 {\it V} $  & 3.96 & 1.11 & 1.20 & 4.23 & 0.303 & 1.06  {\it V} &
8 & {\footnotesize Popper (\cite{Popper97}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize Zwitter et~al. (\cite{Zwitter}) }\\
\hline
 21 & $UU$  &  $0\fd46846016$   &
$\underline{0}$.$\underline{4}$1$\underline{~}$ & 1.40 & 1.41 &
3.35 & 0.444 & 0.97 {\it V} &
$88\fdg5$ & {\footnotesize Yamasaki et~al. (\cite{Yamasaki}) }\\
   & $Lyn$ & $F3 {\it V} + [K7] {\it V}$   & 3.18 & 0.57 & 0.90 & 6.44 & 0.285 & 16   {\it V} &
6 & {\footnotesize Yamasaki et~al. (\cite{Yamasaki86}) }\\
\hline 22 & ${\it V}$361 & $0\fd30961493^{\star}$ & $\underline{0.69}$ &
1.26 & 1.02 & 4.34 & 0.412
& 0.92 {\it V} & $85\fdg7$ & {\footnotesize Hilditch et~al. (\cite{Hilditch}) }\\
   & $Lyr$ & $ F9 {\it V} + [K4] {\it V} $ & 2.48 & 0.87 & 0.72 & 6.46 & 0.289 & 5.66 {\it V} &
7 & \\
\hline 23 & $DD$ & $0\fd56802738^{\star}$ &
$\underline{0}$.$\underline{4}$5$\underline{~}$ & 1.05 & 1.36 &
3.45 & 0.410 & 0.88 {\it V} &
$76\fdg5$ & {\footnotesize Shengbang et~al. (\cite{Shengbang}, \cite{Shengbang97}) }\\
   & $Mon$ & $ F5 {\it V} + [K1] {\it V} $ & 3.32 & 0.47 & 1.01 & 5.29 & 0.304 & 4.03 {\it V} &
5 & {\footnotesize Yamasaki et~al. (\cite{Yamasaki90}) }\\
\hline 24 & $U{\it V}$ & $0\fd861046566^{\star}$ & $\underline{0.78}$ &
0.98 & 1.15 & 4.39 & 0.252
& 0.85 {\it V} & $86\fdg3$ & {\footnotesize Budding et~al. (\cite{Budding96}) }\\
   & $Psc$ &  $ G 5 {\it V} + K 3 {\it V} $  & 4.58 & 0.76 & 0.89 & 5.79 & 0.194 & 3.36 {\it V} &
9 & {\footnotesize Popper (\cite{Popper97}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize  Heckert (\cite{Heckert}) }\\
\hline 25 & $RS$  &  $0\fd664238549$  & $\underline{0.60}$ &
1.36 & 1.58 & 2.95 & 0.381 &
0.90 {\it V} & $87\fdg3$ & {\footnotesize King \&  Hilditch (\cite{King}) }\\
   & $Sct$ &  $ F4 {\it V} + [K2] {\it V}$ & 4.15 & 0.82 & 1.24 & 4.96 & 0.300 & 5.58 {\it V} &
7 & {\footnotesize Buckley (\cite{Buckley}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize Kreiner et~al. (\cite{Kreiner}) }\\
\hline 26 & $GR$  &  $0\fd42985060$   &
$\underline{0}$.$\underline{2}$2$\underline{~}$ & 1.42 & 1.34 &
3.32 & 0.464 & 0.98 {\it V} &
$71\fdg9$ & {\footnotesize Lazaro et~al. (\cite{Lazaro}) }\\
   & $Tau$ & $(F1) {\it V} + [K6] {\it V}$ & 2.88 & 0.31 & 0.76 & 6.72 & 0.264 & 15.8 {\it V} &
4 & {\footnotesize Zhang et~al. (\cite{Zhang}) }\\
\hline 27 & $GY$ & $0\fd4460348^{\star}$ & 0.8: & 1.1  & 1.2  &
3.65 & 0.39  & 0.73 {\it V} &
$88^{\circ}$ & {\footnotesize Lipari \&  Sistero (\cite{Lipari}) }\\
   & $Tel$ & $ F2 {\it V} + [G1] {\it V} $ & 3.1  & 0.9  & 1.0  & 4.60 & 0.32  & 1.8  {\it V} &
1 & {\footnotesize Shaw (\cite{Shaw}) }\\
\hline 28 & $XY$  &  $0\fd47899493$   & $\underline{0.61}$ &
1.10 & 1.13 & 4.42 & 0.363
& 0.97 {\it V} & $81^{\circ}$ & {\footnotesize Pojmanski (\cite{Pojmanski}) }\\
   & $U\!Ma$ &  $ G 2 {\it V} + K 5 {\it V} $  & 3.11 & 0.66 & 0.66 & 7.36 & 0.211 & 12.9 {\it V} &
7 & {\footnotesize Cameron \&  Hilditch (\cite{Cameron97}) }\\
  &  &   &   &  &  &  &  &  &  & {\footnotesize Hilditch \&  Bell (\cite{Hilditch94}) }\\
\hline 29 & $BH$  &  $0\fd81687080$   & $\underline{0.90}$ &
1.16 & 1.22 & 4.03 & 0.255
& 0.63 {\it V} & $86\fdg5$ & {\footnotesize Popper (\cite{Popper97}) }\\
   & ${\it V}ir$ &  $ F 8 {\it V} + G 5 {\it V} $  & 4.79 & 1.05 & 1.12 & 4.52 & 0.233 & 1.42 {\it V} &
7 & {\footnotesize Arevalo et~al. (\cite{Arevalo02}) }\\
\hline 30 & $CG$  &   $0\fd935298$    & $\underline{0.53}$ &
1.43 & 1.64 & 3.2  & 0.314
& 0.90 {\it V} & $78\fdg5$ & {\footnotesize Shen et~al. (\cite{Shen}) }\\
   & ${\it V}ir$ &  $ F6 I{\it V} + K 5 {\it V} $  & 5.22 & 0.76 & 1.50 & 5.0  & 0.287 & 7.52 {\it V} &
6 & \\
\hline 31 & $ER$ & $0\fd69809409^{\star}$ & $\underline{0.96}$ &
1.10 & 1.11 & 4.37 & 0.259
& 0.55 {\it V} & $67\fdg3$ & {\footnotesize Hill et~al. (\cite{Hill}) }\\
   & ${\it V}ul$ &  $ G 0 {\it V} + G 5 {\it V} $  & 4.27 & 1.05 & 1.08 & 4.57 & 0.253 & 1.17 {\it V} &
8 & {\footnotesize Olah et~al. (\cite{Olah}) }\\
  &  &   &   &  &  &  &  &  &  &  {\footnotesize Zeinali \&  Edalati (\cite{Zeinali}  }\\
  &  &   &   &  &  &  &  &  &  &  {\footnotesize Garcia-Sanchez et~al. (\cite{Garcia-Sanchez}) }\\
\hline
\end{tabular}
\end{table}